Author's response to reviews

Title: Educating the Power: HIV/AIDS and Parliamentarians of Pakistan

Authors:

Mohammad A Rai (mohammadalarai@yahoo.com)
Alefiyah Rajabali (alefiyah.r@gmail.com)
Muhammad N Khan (nausherwan2000@hotmail.com)
Mohammad A Khan (mohammad.aslam@aku.edu)
Syed H Ali (syed.ali@aku.edu.pk)

Version: 7 Date: 10 September 2009

Author's response to reviews:

Dear BioMed Central Editorial Production Team,

As requested, the following changes have been brought about:

- the commentary has been organized into the required sections i.e. Abstract, Introduction, Discussion, Conclusions, Authors' contributions, Acknowledgements, References

- the Title page and Abstract from the additional files submitted with last submission have been removed, and now have included these sections in the main manuscript

Thank you very much and do please let us know if there are any further queries.

Regards,

MA Rai